Author's response to reviews

Title: Utility of the Diffusion-weighted Imaging for Activity Evaluation in Crohn’s Disease Patients Underwent Magnetic Resonance Enterography Feng Qi1*, Shen Jun2*, Qiao Yu Qi2, Peng Jiang Chen2, Geng Xiao Chuan1, Zhu Jiong1#, Xu Jian Rong1# *These two authors contribute equally to this manuscript and share co-first authorship #Correspondence and requests for materials should be addressed to Zhu Jiong (billzhu49@hotmail.com) and Xu Jian Rong (xujianrong@renji.com) Feng Qi1*: fengqi_med@126.com Shen Jun2*: shenjun@vip.163.com Qiao Yu Qi2: medqyq@gmail.com Peng Jiang Chen2: pengjc0526@163.com Geng Xiao Chuan1: gengxiaochuan1949@126.com

Authors:

Qi Feng (fengqi_med@126.com)
Jun Shen (shenjun79@sina.cn)
Yuqi Qiao (medqyq@gmail.com)
Jiangchen Peng (pengjc0526@163.com)
Xiaochuan Geng (gengxiaochuan1949@126.com)
Jiong Zhu (billzhu49@hotmail.com)
Jianrong Xu (xujianrong@renji.com)

Version: 7 Date: 21 December 2014

Author's response to reviews: see over
1. I have removed the author emails and corresponding information from the title of my manuscript and formed the part of title page in the manuscript as follow:

Title page

*These two authors contribute equally to this manuscript and share co-first authorship

#Correspondence and requests for materials should be addressed to Zhu Jiong (billzhu49@hotmail.com) and Xu Jian Rong (xujianrong@renji.com)

Feng Qi*: fengqi_med@126.com

Shen Jun2*: shenjun@vip.163.com

Qiao Yu Qi2: medqyq@gmail.com

Peng Jiang Chen2: pengjc0526@163.com

Geng Xiao Chuan1: gengxiaochuan1949@126.com

2. I have rearranged the figure and rewrote the illustration for each figure.